

Article



## Exploring the Impact of Additive Shortcuts in Neural Networks via Information Bottleneck-like Dynamics: From ResNet to Transformer

Zhaoyan Lyu \* D and Miguel R. D. Rodrigues D

Department of Electronic and Electrical Engineering, University College London, London WC1E 6BT, UK; m.rodrigues@ucl.ac.uk

\* Correspondence: z.lyu.17@ucl.ac.uk

Abstract: Deep learning has made significant strides, driving advances in areas like computer vision, natural language processing, and autonomous systems. In this paper, we further investigate the implications of the role of additive shortcut connections, focusing on models such as ResNet, Vision Transformers (ViTs), and MLP-Mixers, given that they are essential in enabling efficient information flow and mitigating optimization challenges such as vanishing gradients. In particular, capitalizing on our recent information bottleneck approach, we analyze how additive shortcuts influence the fitting and compression phases of training, crucial for generalization. We leverage Z-X and Z-Y measures as practical alternatives to mutual information for observing these dynamics in high-dimensional spaces. Our empirical results demonstrate that models with identity shortcuts (ISs) often skip the initial fitting phase and move directly into the compression phase, while non-identity shortcut (NIS) models follow the conventional two-phase process. Furthermore, we explore how IS models are still able to compress effectively, maintaining their generalization capacity despite bypassing the early fitting stages. These findings offer new insights into the dynamics of shortcut connections in neural networks, contributing to the optimization of modern deep learning architectures.

**Keywords:** deep learning; neural networks; transformer; shortcut connections; information bottleneck theory

## 1. Introduction

Machine learning, especially deep learning, has made significant strides in recent years, providing exceptional performance across a variety of tasks and domains [1]. Deep learning has been particularly outstanding in fields like computer vision [2–4], natural language processing [5], and autonomous systems [6], where the capacity to model complex patterns from data has revolutionized traditional approaches [1]. These advancements have been driven by increasingly sophisticated neural network architectures, which continue to evolve to address new challenges [7,8].

One of the central innovations within deep learning architectures is the use of different neural network structures designed to optimize performance and efficiency. From the early multi-layer perceptrons (MLPs) [9] to the more advanced convolutional neural networks (CNNs) [3], recurrent neural networks (RNNs) [10], and attention-based architectures like Transformers [5], the evolution of these models has unlocked new capabilities for machine learning systems.

Among these developments, **additive shortcut connections** have emerged as a fundamental architectural feature that enhances the training of deep neural networks. Introduced to mitigate issues like the vanishing gradient problem, shortcut connections enable the flow of information across layers more efficiently [4]. These connections, particularly in architectures like ResNet [4] and its derivatives, allow neural networks to bypass intermediate layers, facilitating the learning process and improving the performance. Shortcut



Citation: Lyu, Z.; Rodrigues, M.R.D. Exploring the Impact of Additive Shortcuts in Neural Networks via Information Bottleneck-like Dynamics: From ResNet to Transformer. *Entropy* 2024, 26, 974. https://doi.org/ 10.3390/e26110974

Academic Editor: Andrea Murari

Received: 30 October 2024 Revised: 10 November 2024 Accepted: 12 November 2024 Published: 14 November 2024



**Copyright:** © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). structures are now the go-to choice for most state-of-the-art (SOTA) models, especially in vision-based tasks, and are integral to architectures such as Vision Transformers and MLP-Mixers [11,12].

However, despite the widespread success of these models, our understanding of how these shortcut connections influence the training process remains incomplete. While shortcuts help alleviate optimization challenges, they introduce complexities into how neural networks learn and generalize [13]. Specifically, the way networks with shortcut connections navigate the fitting and compression phases commonly observed in the training process—as described by the information bottleneck theory [14]—remains unclear. This gap in understanding is particularly pronounced for networks like Transformers, where the presence of identity shortcuts allows information to bypass layers, potentially altering the traditional pathways to generalization [5]. The mechanisms through which these models retain and compress information, and the effect of shortcut connections on these processes, have yet to be fully explored.

This paper focuses on exploring the role of additive shortcut connections in modern neural networks, particularly their effect on the training dynamics of models such as ResNet and Transformers. Specifically, we aim to investigate how models equipped with these shortcuts behave during the fitting and compression phases of training, which are critical for the model's generalization ability. This exploration is grounded in information bottleneck theory [14], which suggests that neural networks aim to increase the information relevant to the ground-truth labels in their representations while reducing irrelevant information [15]. When examining neural network training dynamics through the information bottleneck framework, it has been found that the training process typically involves two key phases: an initial fitting phase, where relevant information is captured, followed by a compression phase, where irrelevant details are discarded [16].

However, the information bottleneck theory and its dynamics are based on mutual information (MI) measures, which are notoriously difficult to estimate for high-dimensional random variables. Our previous work [17] introduced practical measures for analyzing these phases using Z-X and Z-Y metrics, which are based on the minimal mean squared error (MMSE) and conditional entropy, respectively. These metrics allow for a more reliable analysis of the generalization pathways, circumventing the computational challenges of traditional MI-based approaches. In this study, we leverage this framework to examine how these dynamics are affected by the presence of additive shortcut connections, specifically focusing on architectures such as ResNet, Vision Transformers, and MLP-Mixers.

#### 1.1. Contributions

This paper makes the following key contributions to the study of neural networks with additive shortcut connections, particularly in the context of understanding their training dynamics:

1. **Identification of distinct fitting and compression behaviors:** We demonstrate that models with *non-identity shortcuts* (such as traditional ResNet architectures) follow the conventional two-phase training process, consisting of an initial fitting phase followed by a subsequent compression phase, similar to the behavior seen in traditional feed-forward networks [17].

In contrast, models with *identity shortcuts*, such as Vision Transformers and MLP-Mixers, deviate from this pattern by skipping the initial fitting phase and moving directly into the compression phase. This deviation represents a significant challenge to traditional views of neural network optimization, where both fitting and compression are typically expected phases of training.

2. Analysis of the mechanisms underlying the absence of a fitting phase: We conjecture that models with identity shortcuts are able to skip the fitting phase because the shortcut structure enables the model to propagate all necessary information for the classification task to deeper layers without the need for early-stage fitting.

This conjecture is validated empirically by comparing Z-Y measures at initialization. These comparisons show that models with identity shortcuts retain sufficient information for the task even without requiring an explicit fitting phase, indicating that the network can bypass the typical initial training process.

- 3. Analysis of the mechanisms driving compression: We also explore the mechanisms that drive compression in models with identity shortcuts. Through extensive empirical experiments, we identify two distinct mechanisms:
  - In ResNet-like models with identity shortcuts, compression is caused by the *canceling effect* between the so-called functional and informative components (which we will define later) of the network's representations.
  - In Transformers and MLP-Mixers, compression occurs when the functional component of the network representations *overwhelms* the informative component, leading to a pronounced compression phase.

## 1.2. Organization of the Paper

The paper is organized as follows:

- Section 1 provides an overview of the relevance of deep learning architectures and the importance of additive shortcut connections. It introduces the motivation behind studying the fitting and compression phases in these architectures.
- Section 2 discusses the Transformer architecture, state-of-the-art deep learning models, and shortcut structures in neural networks, with a focus on understanding their generalization and training dynamics.
- Section 3 introduces the Z-X and Z-Y measures adopted in this work, providing a background on their use to observe fitting and compression phases in neural networks.
- Section 4 presents empirical results comparing neural networks with and without shortcut connections, with a particular focus on ResNet, ViT, and MLP-Mixer models. The Z-X dynamics of models with identity and non-identity shortcuts are explored in detail.
- Section 5 analyzes the empirical findings, particularly the reasons why IS models skip the fitting phase and how they manage to compress effectively. This section further explores the interaction between the functional and informative components in network representations.
- Section 6 summarizes the key findings of the paper and outlines potential directions for future research.

## 1.3. Notations

The mathematical notations used in this paper are summarized as follows: We use capital letters such as *X* and *Y* to represent matrices or tensors, where *X* typically denotes input data, and *Y* denotes the corresponding labels or targets. Throughout this paper, we refer to layers in a neural network with subscripted notations, such as  $Z_l$ , where *l* refers to the layer index and *Z* represents the intermediate representation at that layer. The notation  $f_{\theta}$  is used to denote functions or transformations parameterized by a set of learnable parameters  $\theta$  within the neural network.

#### 2. Related Work

## 2.1. The Transformer as a State-of-the-Art Deep Learning Architecture

The Transformer model has rapidly become one of the most influential architectures in the field of deep learning, particularly in natural language processing (NLP) [5]. Its success lies in its ability to capture complex dependencies in data using self-attention mechanisms, which allow the model to attend to different parts of an input sequence or image without the limitations of fixed-size receptive fields.

In recent years, the Transformer architecture has been extended beyond NLP into the domain of computer vision, most notably through the development of the **Vision Transformer** (ViT) [11]. The ViT leverages the same self-attention mechanism to process image patches, treating them as sequences, thereby bypassing the need for convolutional layers. ViT has shown state-of-the-art (SOTA) performance on several vision benchmarks, particularly in classification tasks, where it competes with or even surpasses traditional CNN-based architectures like ResNet [4,18].

Despite their widespread adoption, Transformers also present new challenges in terms of training and generalization [19]. Their high computational demand, reliance on large datasets, and potential susceptibility to overfitting raise important questions about their efficiency and scalability [20]. These challenges make it crucial to further investigate the internal mechanisms of Transformers, especially in how they navigate the fitting and compression phases during training. Understanding these dynamics is particularly important as Transformers incorporate identity shortcuts, which may influence the way they handle information compared to more traditional architectures like CNNs.

#### Understanding Transformers

As the Transformer architecture has risen to prominence, a growing body of research has aimed to understand its inner workings, particularly the mechanisms that enable it to generalize so effectively across tasks and domains. These efforts have primarily focused on explaining how the self-attention mechanism and multi-layer structure contribute to the model's performance and how they differ from more traditional architectures such as CNNs and RNNs [5,21].

One line of research has investigated the self-attention mechanism itself, seeking to explain how it captures long-range dependencies and the role of multi-head attention in learning diverse representations. Studies have shown that attention heads in Transformers often learn different patterns, with some focusing on local relationships while others capture more global dependencies [11,21,22]. This versatility allows Transformers to model complex interactions in the data, but the exact contributions of individual attention heads and layers to the overall learning process remain topics of active investigation.

Another important research direction has focused on the positional encoding component, which provides the necessary ordering information in a model that inherently lacks a sense of sequence [5,23]. Positional encoding is critical for the model's ability to handle sequential data like language or structured inputs like images (in the case of Vision Transformers). Researchers have explored different types of positional encoding—learned versus fixed—and their impact on the model's ability to generalize across different types of data [23–26].

In addition, several studies have attempted to visualize and interpret the internal representations learned by Transformers. Some works have used attention visualization techniques to map out which parts of an input are considered most important by the model [27]. These visualizations help to demystify how Transformers make decisions but also highlight a key challenge: unlike CNNs, where activations can often be linked to specific spatial features, attention maps can be more abstract and harder to interpret in terms of direct feature relationships [28]. This makes understanding the fitting and compression phases of training in Transformers particularly challenging, as their mechanisms for storing and discarding information are less intuitive than those of CNNs.

Finally, the presence of identity shortcuts in Transformers—similar to those found in ResNet—adds another layer of complexity to understanding these models' generalization dynamics. These shortcuts allow information to flow directly to deeper layers without it undergoing transformation in every block, potentially bypassing important fitting processes that occur in other architectures. The question of how these shortcuts impact the compression of information and what role they play in the model's overall training dynamics is a central focus of this paper.

#### 2.2. Exploring Shortcut Structures in Neural Networks

The introduction of shortcut connections into neural networks was a groundbreaking development that aimed to address the challenges of training deep models, particularly the

issue of vanishing gradients. Early neural networks suffered from the inability to propagate gradients effectively across multiple layers, leading to poor optimization, especially in very deep architectures. Shortcut connections, also known as skip connections, were first popularized by the ResNet architecture, which allowed gradients to flow more freely through layers by bypassing intermediate transformations.

In ResNet [4], the key innovation was the use of additive shortcuts, which merge the input of one layer with a deeper layer additively. This architectural modification proved to be crucial in enabling the training of very deep networks with hundreds or even thousands of layers, leading to significantly improved performance on a wide range of tasks. The success of ResNet sparked further exploration into different types of shortcut structures and their implications for network training and performance.

Subsequent research has explored other forms of shortcut connections, such as the concatenation shortcuts used in DenseNet [29]. Unlike the identity shortcuts in ResNet, DenseNet's concatenation shortcuts combine the outputs of all previous layers, allowing each layer to access not only its immediate predecessors but also distant layers' outputs. This creates a more densely connected architecture that encourages feature reuse and leads to more compact models, as fewer parameters are needed to achieve the same performance levels. These concatenation shortcuts have been shown to enhance the learning efficiency of networks, particularly in scenarios with limited data.

Beyond ResNet and DenseNet, modern architectures like Vision Transformers (ViTs) and MLP-Mixers have also incorporated shortcut connections, though in different ways [11,12,30]. In these models, shortcut connections play a critical role in maintaining stability and enabling the effective training of very deep networks. The additive shortcuts in Vision Transformers, for instance, allow information to be passed across layers without any transformation, which we later referred to as identity shortcuts.

While shortcut connections have undeniably eased the training process and improved the performance of deep networks, their role in the fitting–compression dynamic remains an area of active research. Understanding how these connections impact the flow of information through networks, and how they affect the model's ability to fit and compress data during training, is crucial to optimizing these architectures for different tasks.

## 2.3. Methodologies for Understanding Neural Networks

In the effort to demystify the behavior of neural networks, several methodologies have been developed and applied to gain insights into their operations. These methods can be categorized into three primary approaches: explainable deep learning, generalization bound analysis, and network dynamics analysis.

## 2.3.1. Explainable Deep Learning

Explainable deep learning focuses on making neural network decisions more transparent by visualizing the internal representations or learned features at different layers of the model. Early studies in this area predominantly centered on feature visualization, where methods like saliency maps and activation mapping were used to highlight which parts of an input, such as an image, had the greatest influence on the network's prediction [31]. Techniques such as Grad-CAM and its variants have been widely adopted for this purpose, offering a glimpse into which regions of an input are emphasized during inference [27,32–35].

However, these visual explanations are often limited to specific input examples and do not provide a complete understanding of the network's behavior across the entire dataset or in different operational conditions. As a result, while explainability techniques contribute valuable insights into how networks handle individual samples, they are insufficient for analyzing the broader generalization capabilities of the model. This has led to the pursuit of more comprehensive methods for understanding neural networks from a probabilistic or statistical perspective.

#### 2.3.2. Generalization Bound Analysis

Generalization bounds provide a more theoretical approach, using statistical learning theory to describe how well a model trained on a given dataset is likely to perform on unseen data. This method relies on concepts such as the Vapnik–Chervonenkis (VC) dimension and PAC learning to derive the bounds on a model's performance [36–39]. These approaches help quantify the ability of a model to generalize by estimating how much information a model retains from the training data and how that affects its performance on new data.

While generalization theory has yielded important insights, there are several limitations to its practical application. For instance, deriving tight bounds for deep neural networks, especially large models with many layers, remains a challenge [40]. Additionally, while generalization bounds can describe the potential behavior of a model, they do not fully capture the dynamic process of training, particularly how models transition from fitting the data to effectively generalizing across unseen instances.

#### 2.3.3. Information Bottlenecks and Network Dynamics Analysis

The information bottleneck (IB) theory has emerged as a powerful framework for understanding the learning dynamics of neural networks, providing insights into how models generalize by balancing the retention of relevant information (fitting) and the discarding of irrelevant details (compression) [14,15]. Initially developed in the context of information theory, the IB principle posits that neural network learning can be conceptualized as a process of capturing and retaining only essential information for the task at hand while compressing redundant or irrelevant features [14]. This two-phase dynamic, marked by an initial fitting phase followed by a compression phase, is crucial to understanding how neural networks transition from learning to generalizing on unseen data [15].

A growing body of research has used the IB framework to analyze how information flows through layers during training. A notable study by Tishby et al. [16] introduced mutual information as a tool for tracking the information flow within neural networks, illustrating that they generally exhibit an early period of fitting, where useful information is retained, followed by an extended phase of compression, where unnecessary details are discarded. This framework has provided theoretical support for why deeper networks often generalize better despite their higher capacity to overfit, suggesting that effective compression enhances their generalization capabilities. However, estimating mutual information in high-dimensional spaces, as required for complex networks with many parameters and layers, poses significant challenges [41–43]. These challenges have led to varying findings in the literature, with some studies observing strong compression phases and others reporting minimal or even absent compression [13,16,44–47].

To address these limitations, recent work has proposed more computationally feasible alternatives to mutual information, such as MMSE-based Z-X measures and conditionalentropy-based Z-Y measures [17]. These metrics offer a practical way to observe the network dynamics without the complexities associated with mutual information estimation. Studies leveraging Z-X and Z-Y measures have reaffirmed the presence of fitting and compression phases in simpler architectures like feed-forward neural networks while also extending this analysis to complex structures such as CNNs and MLPs.

Building on these foundations, this paper explores how additive shortcut connections in architectures like ResNet and Transformers influence the generalization dynamics. While the IB theory has helped reveal how deep networks generally manage information retention and compression, the impact of shortcut connections on these processes is less understood. By extending IB analysis to models with additive shortcuts, this work aims to clarify how these connections affect the training dynamics, particularly in modern architectures such as Vision Transformers and MLP-Mixers. Additionally, the use of Z-X and Z-Y measures offers a practical framework for overcoming some of the computational challenges associated with traditional IB approaches, facilitating a more detailed examination of how shortcut connections navigate the fitting and compression phases to improve generalization [17].

## 3. Recap: The Z-X Measure and the Z-Y Measure for Observing the Fitting and Compression Phases

In this paper, we adopt the Z-X and Z-Y measures proposed in our previous work [17] to study the dynamics of neural networks with additive shortcuts. As defined in [17], we refer to the networks under investigation (such as CNNs, Transformers, and MLP-Mixers) as the **subject networks**. This section briefly introduces the definitions of Z-X and Z-Y measures and their associated dynamics.

We focus on image classification problems, using *X* to represent the image input and *Y* as the ground-truth labels. The intermediate representation of layer *l* is denoted as  $Z_l$ . Formally, we model the classification with a Markov chain as follows:

$$Y \to X \to Z_1 \to \cdots \to Z_l \to \cdots \to Z_L$$

The representation at the last layer  $Z_L$  corresponds to the network's prediction.

In this Markov chain structure, Y is positioned at the beginning because it represents the target label information that flows through the network. This setup reflects the progression of information from the ground-truth label Y to the input X and subsequently through each layer's representation  $Z_l$ . Placing Y at the start of the chain is essential in the IB theory and related analyses, as it ensures that  $Z_L$  (and all intermediate representations  $Z_l$ ) encodes information about Y derived through X. This Markov chain structure captures the dependency relationships needed for IB analysis, where the information learned by  $Z_l$ is traceable to Y via X, reinforcing that each layer's representation gradually refines the information needed for classification.

#### 3.1. The Z-X Measure and the Z-Y Measure

In our previous work, we introduced two key measures—Z-X and Z-Y measures—to provide more reliable insights into neural network dynamics, specifically focusing on the **fitting** and **compression phases**. These measures serve as alternatives to mutual information (MI), which is difficult to estimate in high-dimensional settings. The Z-X and Z-Y measures are computationally feasible.

#### 3.1.1. The Z-X Measure

The Z-X measure is the MMSE (minimum mean squared error) between the neural network representation *Z* and the input data *X*, estimated by a neural-network-based estimator in a data-driven manner. More specifically, given a dataset  $\mathcal{D} = (X_1, Y_1), \ldots, (X_{|\mathcal{D}|}, Y_{|\mathcal{D}|})$ , we estimate the Z-X measure  $m_{Z:X}$  as follows:

$$\begin{split} m_{Z;X} &= \text{mmse}(Z|X) \\ &\approx \min_{\phi} \frac{1}{|\mathcal{D}|} \sum_{k=1}^{|\mathcal{D}|} (f_{\phi}(Z_{l;k}) - X_{k})^{\top} (f_{\phi}(Z_{l;k}) - X_{k}) \\ &= \frac{1}{|\mathcal{D}|} \sum_{k=1}^{|\mathcal{D}|} (f_{\phi^{*}}(Z_{l;k}) - X_{k})^{\top} (f_{\phi^{*}}(Z_{l;k}) - X_{k}) \\ &= \frac{1}{|\mathcal{D}|} \sum_{k=1}^{|\mathcal{D}|} (f_{\phi^{*}} \circ f_{\theta_{1:l}}(X_{k}) - X_{k})^{\top} (f_{\phi^{*}} \circ f_{\theta_{1:l}}(X_{k}) - X_{k}) , \end{split}$$
(1)

where  $f_{\theta_1}(X_k)$  represents the subject network's forward pass from layer 1 to layer *l* for the input sample  $X_k$ , and  $Z_{l;k}$  is the corresponding representation generated by the subject network at layer *l* for the input sample  $X_k$ .  $f_{\phi}$  is a separate estimator network, parameterized by  $\phi$ , trained to reconstruct the input  $X_k$  from the intermediate representation of the subject network  $Z_{l;k}$ . The Z-X measure quantifies how much of the original input information is retained by the network at layer *l*.

## 3.1.2. The Z-Y Measure

The Z-Y measure captures how well the neural network representation  $Z_l$  at layer l can predict the target label Y. It is defined as the conditional entropy H(Y|Z), which quantifies the uncertainty in predicting label Y given representation  $Z_l$ . The Z-Y measure is estimated similarly to the Z-X measure, using a neural-network-based predictor  $f_{\psi}(Z_{l;k})$  for the label  $Y_k$ . Specifically, it is defined as follows:

$$m_{Z;Y} = H(Y|Z)$$

$$\approx \min_{\psi} \frac{1}{|\mathcal{D}|} \sum_{k=1}^{|\mathcal{D}|} \ell_{CE}(f_{\psi}(Z_{l;k}), Y_{k})$$

$$= \frac{1}{|\mathcal{D}|} \sum_{k=1}^{|\mathcal{D}|} \ell_{CE}(f_{\psi^{*}}(Z_{l;k}), Y_{k})$$

$$= \frac{1}{|\mathcal{D}|} \sum_{k=1}^{|\mathcal{D}|} \ell_{CE}(f_{\psi^{*}} \circ f_{\theta_{1:l}}(X_{k}), Y_{k}),$$
(2)

where  $f_{\psi}$  is a separate estimator network parameterized by  $\psi$ , trained to predict the label  $Y_k$  from the internal representation  $Z_{l;k}$ .  $\ell_{CE}$  represents the cross-entropy loss function.

#### 3.2. Estimating the Z-X and Z-Y Measures

The estimation of the Z-X and Z-Y measures involves training separate neural networks to minimize the squared loss (for the Z-X measure) and the cross-entropy loss (for the Z-Y measure). These processes are shown in Figure 1 and can be described as follows:

**Z-X estimation:** For each input  $X_k$ , the subject network's forward pass generates the representation  $Z_{l;k}$  at layer l. This representation is passed through the Z-X estimator network  $f_{\phi}$ , which is trained to reconstruct  $X_k$  by minimizing the mean squared error (MSE) between  $f_{\phi}(Z_{l;k})$  and the original input  $X_k$ , as shown in Equation (1).

**Z-Y estimation:** Similarly, to estimate the Z-Y measure, the representation  $Z_{l;k}$  is passed through the Z-Y estimator network  $f_{\psi}$ , which is trained to predict the target label  $Y_k$  by minimizing the cross-entropy loss, as shown in Equation (2).



**Figure 1.** The framework for estimating the Z-X and Z-Y measures. This figure is adapted from Figure 1 in [17] .  $\ell_{SE}$  refers to the squared loss, and  $\ell_{CE}$  represents the cross-entropy loss.

To address potential overfitting, we split the dataset  $\mathcal{D}$  used for Z-X and Z-Y estimation into two subsets:  $\mathcal{D}_{train}$  and  $\mathcal{D}_{valid}$  (Unless otherwise specified, we split the dataset  $\mathcal{D}$  into  $\mathcal{D}_{train}$  and  $\mathcal{D}_{valid}$  with a 7:3 ratio in this study). The training subset  $\mathcal{D}_{train}$  is used to train the estimator networks, while the validation subset  $\mathcal{D}_{valid}$  ensures that the estimators do not overfit. The final Z-X and Z-Y measures are computed based on the validation set, ensuring robust estimation. Both estimators are trained using a gradient-descent-based algorithms until convergence, identified when the estimated measures on the validation set cease to decrease. At this point, the final Z-X and Z-Y measures are obtained based on the validation set for each layer *l* of the subject network. The complete training procedure for estimating the Z-X and Z-Y measures is outlined in Algorithm 1.

**Algorithm 1:** Estimate the Z-X measure and the Z-Y measure with the validation set.

**Input:**  $f_{\theta}$  the subject network at checkpoint *t*, random seed *s*, patience  $\tau \in \mathcal{N}_+$ **Data:**  $\mathcal{D} = \{(X_k, Y_k)\}_{k=1}^{|\mathcal{D}|}$ **Output:**  $\{m_{Z_1;X}, \ldots, m_{Z_L;X}\}, \{m_{Z_1;Y}, \ldots, m_{Z_L;Y}\}$ Split  $\mathcal{D}$  into  $\mathcal{D}_{\text{train}}$  and  $\mathcal{D}_{\text{valid}}$ ; Obtain subject network representations  $\{\{Z_{l,k}\}_{k=1}^{|\mathcal{D}|}\}_{l=1}^{L} \leftarrow \{f_{\theta}(\{X_k\}_{k=1}^{|\mathcal{D}|})\}_{l=1}^{L};$  $l \leftarrow 1;$ while  $l \leq L$  do Initialize  $f_{\phi_l}$ ,  $f_{\psi_l}$  with random seed *s*;  $p \leftarrow 0 \ m^* \leftarrow \infty$ ,  $update_flag \leftarrow$  True; while  $p < \tau$ ; // Estimating the Z-X measure. do if update\_flag then Train  $f_{\phi_l}$  on  $\mathcal{D}_{\text{train}}$  to minimize  $m_{Z_l;X} = \frac{1}{|\mathcal{D}_{\text{train}}|} \sum_{k=1}^{|\mathcal{D}_{\text{train}}|} \ell_x(f_{\phi_l}(Z_{l,k});X_k);$ Evaluate  $m_{Z_l;X}$  on  $\mathcal{D}_{\text{valid}}$ ; if Validation  $m_{Z_1;X} < m^*$  then  $m^* \leftarrow \text{Validation } m_{Z_l;X}, p \leftarrow 0;$ end else  $p \leftarrow p+1;$ end end if  $p \ge \tau$  then *update\_flag*  $\leftarrow$  False; end end  $p \leftarrow 0, m^* \leftarrow \infty, update\_flag \leftarrow True;$ while  $p < \tau$ ; // Estimating the Z-Y measure. do if update\_flag then Train  $f_{\psi_l}$  on  $\mathcal{D}_{\text{train}}$  to minimize  $m_{Z_l;Y} = \frac{1}{|\mathcal{D}_{\text{train}}|} \sum_{k=1}^{|\mathcal{D}_{\text{train}}|} \ell_y(f_{\psi_l}(Z_{l,k});Y_k);$ Evaluate  $m_{Z_l;Y}$  on  $\mathcal{D}_{\text{valid}}$ ; if Validation  $m_{Z_l;Y} < m^*$  then  $m^* \leftarrow \text{Validation } m_{Z_l;Y}, p \leftarrow 0;$ end else  $| p \leftarrow p+1;$ end end if  $p \ge \tau$  then  $update_flag \leftarrow False;$ end end end

Specific details regarding the estimator network architecture and training setups will be provided in the relevant sections.

#### 3.3. Z-X Dynamics and Z-Y Dynamics

By plotting the Z-X and Z-Y measures for each layer across different stages of training, we can gain valuable insights into how the subject network's internal representations evolve during training. In doing so, at various checkpoints during the training of the subject network, we freeze the parameters of the subject network. We estimate the Z-X and/or Z-Y measures for the current state of the network. Each checkpoint provides a point in the Z-X/Z-Y space, representing the network's ability to retain input information (with the Z-X measure) and predict labels (with the Z-Y measure) during training.

Tracking the Z-X and Z-Y measures over multiple checkpoints allows us to plot the **Z-X dynamics** and **Z-Y dynamics** of the subject network. This will reveal how the network transitions between the fitting and compression phases.

### 3.4. The Fitting Phase and the Compression Phase

As highlighted in our previous work [17], neural networks generally undergo two distinct phases during training: the **fitting phase** and the **compression phase**.

## 3.4.1. The Fitting Phase

In the early part of training, the network typically undergoes a fitting phase, where it captures as much information about the input data as possible. During this phase, the Z-X measure decreases as the network's intermediate layers are learning to retain more information about the input *X*, while the Z-Y measure decreases as the network learns to predict the labels effectively.

#### 3.4.2. The Compression Phase

After the fitting phase, the network enters the compression phase, where it discards unnecessary information. This phase is marked by an increase in the Z-X measure, indicating reduced retention of input information, while the Z-Y measure stabilizes as the network retains the most relevant features for accurate label prediction.

The presence of these two phases during neural network training, as theorized by the information bottleneck theory, has been a topic of much discussion and debate [13,16,44–47]. In this work, we extend the exploration of these phases, focusing specifically on neural networks with additive shortcuts.

## 4. The Fitting and Compression Phases in Models with Additive Shortcuts

#### 4.1. Identity Shortcuts and Non-Identity Shortcuts

We categorize additive shortcuts into two types: identity shortcuts (ISs) and nonidentity shortcuts (NISs). Figure 2 illustrates the structure of both types.



Identity Shortcut (IS)

Non-Identity Shortcut (NIS)

**Figure 2.** An illustration of identity and non-identity shortcuts. The representations at different stages are labeled in pink.

**Identity shortcuts (ISs)** refer to connections where the input is passed directly to a deeper layer without any transformation. These shortcuts "skip" layers, allowing the input to flow through the network unmodified, preserving information across multiple layers.

ISs can only be used in models where the input and output dimensions of a layer are the same, enabling seamless addition to the output. ISs are fundamental in architectures such as Transformers and MLP-Mixers. Mathematically, an IS can be expressed as

$$Z_{l} = f(Z_{l-1}) + Z_{l-1} = Z'_{l} + Z_{l-1},$$

where  $f(Z_{l-1})$  represents the transformation in the main path.

**Non-identity shortcuts (NISs)**, in contrast, apply transformations to the input before passing it to deeper layers. Such transformations, like convolutions or linear operations, are often necessary when the input and output are in different dimensions, as seen in models like ResNet [4]. For instance, in ResNet, NISs are used when downsampling layers reduce the spatial dimensions. In these cases, a convolutional transformation ensures the shortcut can be added to the main path's output without dimension mismatches.

Mathematically, a NIS can be written as

$$Z_{l} = f(Z_{l-1}) + g(Z_{l-1}) = Z'_{l} + Z''_{l}.$$

where  $g(Z_{l-1})$  is the transformation applied to the input from layer l - 1, and  $f(Z_{l-1})$  is the main path's transformation.

In this section, we examine and compare the Z-X dynamics of five neural networks:

- (NS) CNN: A VGG-like CNN serves as the control group with no shortcuts (NS).
- (NIS) **ResCNN**: A ResCNN model incorporating NISs, similar to ResNet [4].
- (IS) iResCNN: A modified ResCNN, incorporating ISs throughout.
- (IS) Vision Transformer (ViT): A ViT model that uses ISs by design.
- (IS) MLP-Mixer: A MLP-Mixer model also with ISs by design.

4.2. The CNN and ResCNN

4.2.1. Setup

**Subject network setup**: We begin by analyzing the Z-X dynamics of a CNN and a residual CNN (ResCNN). Figure 3 (left and middle) illustrates the architectures. This experiment aims to compare the dynamics of networks without shortcuts (NS) and with NISs, laying the foundation for analyzing models with ISs.

As shown in Figure 3, both models feature three modules, each doubling the depth while halving the representation shape. The ResCNN's main path shares the CNN's architecture, with each module adopting a NIS. The shortcut in the ResCNN uses a convolutional layer (with a kernel size of 1) to align the input's shape, allowing for additive merging.

We investigate the Z-X dynamics of  $Z_1$ ,  $Z_2$ , and  $Z_3$  in the CNN and the corresponding  $Z_l$  layers in the ResCNN.

Both models are trained on the CIFAR-10 dataset [48] (The information bottleneck theory has traditionally faced difficulties in estimating mutual information for highdimensional representations, leading many studies to rely on synthetic data [16] or simpler datasets such as MNIST [49]. In contrast, the MMSE-based Z-X measure and conditionalentropy-based Z-Y measure adopted in this paper are more computationally feasible, allowing us to work with higher-dimensional data.) for 50 epochs, with a batch size of 256. The training uses cross-entropy loss with the AdamW optimizer, with a weight decay of 1.0 and a learning rate of 0.0001. No data augmentation is applied. This configuration ensures an efficient training process without pronounced overfitting.

**Z-X estimator setup**: To estimate Z-X measures, the estimator must map the representation back to the input image space. We use the design shown in Figure 4. This setup ensures stable and swift estimation of the Z-X measures.

To track the Z-X dynamics, we freeze the subject network's parameters at various checkpoints, feed the full CIFAR-10 training set through the network to obtain representations, and then optimize the Z-X estimator. The Z-X estimator is trained using the Adam optimizer with a learning rate of 0.0001 for up to 200 epochs, with early stopping after 10 epochs of no test loss improvement.



**Figure 3.** Architecture of the CNN (**left**), ResCNN (**middle**), and iResCNN (**right**). "Conv" refers to convolutional layers, "ReLU" to rectified linear unit activation, and "FC" to fully connected layers. The convolutional kernel and weight matrix shapes are noted in gray, and the tensor/matrix/vector shapes are labeled in blue.



**Figure 4.** Z-X estimator design for convolutional networks. "TConv" stands for transposed convolution, used to upscale feature maps, and "tanh" represents the hyperbolic tangent activation function.

## 4.2.2. The Z-X Dynamics of the CNN and ResCNN

The Z-X dynamics of the CNN and ResCNN are shown in Figure 5 (left and middle). Both networks exhibit a fitting phase (where Z-X measures decrease), followed by a compression phase (where the Z-X measures increase), consistent with previous observations in feed-forward neural networks [17].



**Figure 5.** The Z-X dynamics of the CNN (**left**), ResCNN (**middle**), and iResCNN (**right**). The Z-X measures are estimated at corresponding modules in Figure 3.

#### 4.3. The iResCNN

#### 4.3.1. Setup

**Subject network setup**: The architecture of the iResCNN is detailed as follows: Firstly, the number of channels is increased from 3 to 64 by cyclically repeating the red, green, and blue channels. This approach ensures that there is no information loss when increasing the dimensional representation at the beginning of the neural network. Subsequently, within each residual module, the number of channels and the feature map size remain unchanged, allowing for the implementation of identity shortcuts, as the feature maps consistently maintain a  $64 \times 32 \times 32$  dimension. Each residual block includes two convolutional layers, each succeeded by a tanh activation function. The feature maps are eventually flattened, followed by fully connected layers functioning as the classification head, similar to the ResCNN. The precise architecture of the iResCNN is depicted in the right panel in Figure 3.

The iResCNN is trained and validated on the standard CIFAR-10 dataset, with all the training hyper-parameters mirroring those used for the CNN and ResCNN.

**Z-X estimator setup**: The architecture of the estimator network for the Z-X dynamics and the training setups for the estimator adhere to the configuration established for the ResCNN.

## 4.3.2. The Z-X Dynamics of the iResCNN

The right panel in Figure 5 displays the Z-X dynamics of  $Z_l$  from various residual modules in the specially designed iResCNN.

Notably, this configuration of the residual network largely skips the initial fitting phase, launching directly with a compression phase. In comparison with the control group—the Z-X dynamics of the CNN and ResCNN presented in the left and middle panels in Figure 5—it is safe to conjecture that this behavior is primarily due to the inclusion of ISs within the network.

Following the initial compression, a decline in the Z-X measure in  $Z_l$  of modules 2 and 3 is observed. This subsequent fitting phase, however, coincides with the network exhibiting mild signs of overfitting, as indicated by an increase in the validation set loss.

Overall, during the decrease in the subject network's validation loss, replacing the NISs in the ResCNN with ISs appears to result in Z-X dynamics that bypass the initial fitting phase and commence with a compression phase.

## 4.4. ViTs and MLP-Mixers

ViTs and MLP-Mixers diverge significantly in their structure from CNNs and MLPs. Both of them initially tokenize the input data and then process them through a series of modules featuring ISs. In particular, the ViTs are composed of a sequence of MHSA and FF modules, while MLP-Mixers are composed of a sequence of token-mixers and channel-mixers. The representation maintains a consistent shape throughout the ViTs and MLP-Mixers, up to the classification head, comprising a tensor with n tokens, each of d dimensions. Hence, the additive shortcuts applied to the MHSA, FF, token-mixer and channel-mixer modules in ViTs and MLP-Mixers qualify as ISs.

#### 4.4.1. Setup

**Subject network setup**: The ViT implemented is constituted of six Transformer modules, and the architecture is shown in the left panel in Figure 6.



**Figure 6.** Architecture of ViT (**left**) and MLP-Mixer (**right**). "MHSA" represents multi-head self attention modules, "FF" indicates feed-forward modules, and "GAP" represents global average pooling layers.

The model first segments the image input into patches (64 patches per image,  $4 \times 4$  pixels per patch), which are subsequently embedded into tokens (each with a length of 512) through a learnable embedding module. Additive sine–cosine positional encoding is then utilized. Each MHSA module accommodates eight heads and maintains a dropout rate of 0.1. Note that the ViT does not use a class token. Instead, it adopts global average pooling (GAP) to reduce the token dimensions before feeding it to a linear classification head, as implemented in [18].

The MLP-Mixer—as shown in the right panel in Figure 6—employs the same tokenization as the ViT model and contains six token-mixer and channel-mixer modules, with the token dimensions also fixed at 512. The token-mixer and channel-mixer setups are aligned with the original implementation in [12].

Both the ViT and the MLP-Mixer are trained from scratch on the standard CIFAR-10 dataset using the same recipe: Using the Adam optimizer, the learning rate is initially set to 0.001 and is gradually reduced to 0 following a cosine scheduling pattern [50]. The training period spans 500 epochs, and the batch size is configured to 2048. To augment the training data, an initial application of random augmentation [51] is employed, with the number of operations set to 2 and the magnitude at 14. This is followed by padding

the image with 4 pixels on each side and randomly cropping it back to the original size  $(32 \times 32)$ . A random horizontal flip is also performed. For regularization, we implement Mixup regularization [51] with a hyper-parameter of 0.2, CutMix regularization [52] is also set at a hyper-parameter of 0.2, and label smoothing [53] is used with a hyper-parameter of 0.3. The weight decay parameter is set to 0.001.

This training setup enables the ViT and MLP-Mixer models to be trained from scratch on the CIFAR-10 dataset, achieving performance levels that approach the state-of-the-art benchmarks [54]. Specifically, our ViT model reached an accuracy of 86.14%, while the MLP-Mixer achieved 83.76%, demonstrating effective training and meaningful representation learning.

**Z-X estimator setup**: Compared with the subject networks previously examined that use convolutional layers as the main building blocks, the ViT and MLP-Mixer architectures are more memory-intensive in their implementation [11]. The tokenized representations in the ViT, with a dimensionality of  $64 \times 512 = 32,768$ , require full matrix multiplications for each pairwise token interaction. This contrasts with convolutional networks, which are more memory-efficient due to their ability to use shared kernels across spatial locations, reducing the memory intensity and allowing for dimensionality reduction through operations like pooling.

As a result, it is crucial to design the estimators for measuring the Z-X dynamics of ViTs and MLP-Mixers to be lightweight. Otherwise, it would be challenging to efficiently implement both the ViT and MLP-Mixer subject networks and the associated estimators on a server to estimate the Z-X measures effectively.

As depicted in Figure 7, the Z-X estimator for the ViT and MLP-Mixer begins with reordering of the tokens. Since each token in the ViT and MLP-Mixer representation is initially flattened and embedded from an image patch, this approach intuitively aligns each token with its corresponding image patch. In particular, the tokenize module in the ViT and MLP-Mixer subject networks divides the square CIFAR-10 input image into 64 patches, each embedded into vectors with a length of 512. Consequently, the reordered representation has a shape of  $1 \times 1 \times 8 \times 8 \times 512$ .



Z-X estiamtor for tokenized input

**Figure 7.** Architecture of Z-X estimators for token-based models. For the tokenized representation of the ViT and MLP-Mixer in this paper, n = 64, p = 8, and d = 512.

This step is followed by the application of a transposed convolution operation [55], designed to increase the height and width of the representation to match those of the input data. More specifically, for the ViT and MLP-Mixer models trained on the CIFAR-10 dataset, the initial patch size is  $4 \times 4$ . Therefore, the output of the transposed convolution operation is shaped as  $4 \times 4 \times 8 \times 8 \times 512$ .

Then, a tanh activation function is applied, followed by a convolutional layer aiming to adjust the channel of the representation to ensure the output of the estimator matches the original input data. This design considers the shape of the representation, making the estimator network lightweight yet stable for optimization.

## 4.4.2. The Z-X Dynamics of the ViT and MLP-Mixer

It can be observed from the upper panels in Figure 8 that the ViT, when trained from scratch, also skips the initial fitting phase, commencing instead with an immediate compression phase. Following this, some of the layers experience a mild fitting phase, subsequently leading into another phase of compression observable across all layers, characterized by a slow increase in the Z-X measure. Notably, the Z-X measure corresponding to the output of



the first Transformer module remains near zero throughout the duration of training. This stability can be attributed to its direct connection to the input embeddings via an IS.

**Figure 8.** The Z-X dynamics of the ViT (**top**) and MLP-Mixer (**bottom**). The Z-X measures are estimated at the corresponding modules in Figure 6.

When compared with the iResCNN (the right panel in Figure 5), which is also trained on the CIFAR-10 dataset, the Z-X dynamics of the ViT begin from a comparable value at epoch 0 but exhibit a much more aggressive increase than those of the iResCNN.

In the lower panels of Figure 8, the findings within the MLP-Mixer parallel those noted in the ViT, where the network demonstrates an initial compression phase rather than a fitting phase. Following this initial compression, the early layers transition into a fitting phase, whereas the deeper layers persist in their compression efforts. Both channel-mixers and token-mixers follow similar trends.

Overall, models with ISs generally skip the initial fitting phase and instead exhibit a pronounced compression phase, distinguishing them from neural networks that do not implement shortcuts or that use NISs. Meanwhile, we observe that while ViT models display a milder fitting phase, MLP-Mixer models exhibit a more pronounced fitting phase in their shallower layers. Although the cause of this difference is not entirely clear, we have documented it here for transparency. Despite these variations, all models with ISs retain the ability to compress, suggesting that compression remains essential for generalization in neural networks with ISs.

# 5. Why Does the Behavior of IS-Based Models Differ from the Behavior of NIS-Based Models?

The investigation of the Z-X dynamics in models with additive shortcuts—particularly those with identity shortcuts (ISs)—has highlighted two distinct phenomena: (1) models with ISs tend to skip the initial fitting phase during training, and (2) models with ISs still exhibit pronounced compression phases.

This section further explores these findings by addressing two key questions: (1) why do models with ISs skip the initial fitting phase, and (2) how are models with ISs able to compress effectively despite skipping the fitting phase?

#### 5.1. On the Absence of the Initial Fitting Phase

We conjecture that the absence of the initial fitting phase is due to the compression of the nature of the IS: at random initialization, the model featuring an IS is able to pass almost all the information necessary for the classification task to the deeper layers. This means the network does not need an explicit fitting phase to help the representation in the deep layers to gain information about the ground-truth labels and can start the compression phase directly to achieve generalization.

We designed an experiment to empirically validate our conjecture. The information captured by the representations about the ground-truth labels can be quantified using the Z-Y measure, which is estimated by minimizing Equation (2). If the Z-Y measure is sufficiently low at the random initialization stage (i.e., before any training), we can safely claim that our conjecture is validated.

But what constitutes a "sufficiently" low Z-Y measure? Fortunately, our Z-Y measure is based on minimizing the cross-entropy loss between the estimator's output and the ground-truth labels. This is directly comparable to the loss function used by the subject network in the classification task—the subject networks are trained to minimize the crossentropy loss between their predictions and the ground-truth labels. Hence, we use the lowest cross-entropy loss of the subject network, denoted as  $\ell_{\theta^*}(X;Y)$ , as the threshold for judging the Z-Y measure. If the Z-Y measure,  $m_{Z_l;Y}$ , is comparable to or lower than  $\ell_{\theta^*}(X;Y)$ , we assert that the representation has gained sufficient information about the ground-truth labels.

#### 5.1.1. Estimating Z-Y Measures

We estimate the Z-Y measures for the models at random initialization (i.e., checkpoint 0), with a particular focus on the representations from the deeper layers (closer to the output). According to the data processing inequality, if the deeper layers possess sufficient information about the ground-truth labels, the earlier layers should as well.

The Z-Y estimators for the CNN, ResCNN, and iResCNN share similar architectures. They use the ResCNN architecture shown in the middle panel in Figure 3, with the first convolutional layer adapting to the representation's channel dimension. For the ViT and MLP-Mixer, we employ the ViT model (without the tokenization layer) directly as the Z-Y estimator.

Regarding the optimization setup, the Z-Y estimators for the CNN, ResCNN, and iResCNN mirror the training setups (in terms of the choice of optimizer, learning rate, regularization, and data augmentation) used for the ResCNN subject network. Similarly, the Z-Y estimators for the ViT and MLP-Mixer follow the same setup as those used for training the ViT subject network.

## 5.1.2. Results

Table 1 presents the Z-Y measures for various subject networks at random initialization. The results support our conjecture. As shown in Table 1, the models without ISs (i.e., the CNN and ResCNN) show Z-Y measures higher than  $\ell_{\theta^*}(X;Y)$ . In contrast, the models with ISs (the iResCNN, ViT, and MLP-Mixer) exhibit Z-Y measures lower than  $\ell_{\theta^*}(X;Y)$ . This empirical validation confirms our conjecture that ISs can propagate sufficient information about the ground-truth labels at random initialization. **Table 1.** Z-Y measures compared with the best cross-entropy loss in the subject network.  $Z_{-1}$  and  $Z_{-2}$  refer to the representations from the last module and the second to last module, respectively. Differences between the Z-Y measure and the lowest subject network loss are shown in red (higher) and green (lower).

	CNN		ResCNN		iResCNN		ViT		MLP-Mixer	
$\ell_{\theta^*}(X;Y)$	1.082		1.049		1.393		0.479		0.571	
$m_{Z_{-2};Y}$	1.354	+0.272	1.252	+0.203	1.376	-0.017	0.326	-0.153	0.446	-0.125
$m_{Z-1};Y$	1.470	+0.388	1.313	+0.264	1.380	-0.013	0.328	-0.151	0.448	-0.123

#### 5.2. The Mechanisms Enabling Compression in Models with ISs

As previously outlined, the models equipped with ISs throughout are able to transmit information from the input data to deeper layers easily. However, pronounced compression behaviors are evident in the representations generated by these models. Therefore, this section delves into the composition of these representations, identifying and discussing various scenarios that might lead to compression behavior.

To begin, we propose decomposing the representations into two components—which are the *functional component* and the *informative component*—in order to understand their interaction. We then propose several element-wise statistics to investigate the interplay between the two components and to try to understand the mechanisms behind the compression phenomena of the models featuring ISs.

#### 5.2.1. Methodology

**Decomposition of the representation of a sub-network with ISs**: Let us formulate a module with ISs as follows:

$$Z_{l} = f_{\theta_{l}}(Z_{l-1})$$
  
=  $h_{\theta_{l}}(Z_{l-1}) + Z_{l-1}$  (3)

$$= Z'_l + Z_{l-1}.$$
 (4)

Consider a sub-network represented as  $f_{\theta_{1:l}} = f_{\theta_l} \circ \cdots \circ f_{\theta_1}$ , where each module features identity shortcuts, as formulated in Equations (3) and (4). The representation  $Z_l$  can be decomposed in the following manner:

$$Z_{l} = f_{\theta_{1:l}}(I(X))$$

$$= f_{\theta_{l}} \circ f_{\theta_{l-1}} \circ \cdots \circ f_{\theta_{1}}(I(X))$$

$$= h_{\theta_{l}} \circ f_{\theta_{l-1}} \circ \cdots \circ f_{\theta_{1}}(I(X)) + f_{\theta_{l-1}} \circ \cdots \circ f_{\theta_{1}}(I(X))$$

$$= h_{\theta_{l}} \circ f_{\theta_{l-1}} \circ \cdots \circ f_{\theta_{1}}(I(X)) + h_{\theta_{l-1}} \circ f_{\theta_{l-2}} \circ \cdots \circ f_{\theta_{1}}(I(X)) + \cdots + h_{\theta_{1}}(I(X)) + I(X)$$

$$= \sum_{i=1}^{l} h_{\theta_{i}}(Z_{i-1}) + I(X)$$

$$= \sum_{i=1}^{l} Z'_{i} + Z_{I}$$
(6)

$$=Z_{I\cdot F}+Z_{I},\tag{7}$$

where  $Z_0 = Z_I$ . I(X) is a pre-processing function that applies to the input *X* and does not reduce any information in *X*. For example, the tokenization layers in the Transformers can be viewed as one such function.

In Equation (7), the representation  $Z_l$  is effectively decomposed into two components:  $Z_{l;F}$ , the *functional component*, which is the sum of all  $Z'_l$  processed by the respective residual branches in the residual modules, and  $Z_l$ , the *informative component*, which is a propagation of the pre-processed input I(X).

We now define several element-wise measures to help us investigate the interactions between the functional component and the informative component.

**Element-wise mean and variance**: Consider three random tensors X, Y, and  $Z \in \mathcal{R}^{h \times w \times c}$ , with Z = X + Y. We use  $X_{i,j,k}$ ,  $Y_{i,j,k}$ , and  $Z_{i,j,k}$  to denote specific elements in the three-dimensional tensors X, Y and Z, respectively.

Each element in a tensor, such as  $X_{i,j,k}$ , can be treated as a scalar random variable. The variance in this scalar, denoted as  $\sigma^2_{X_{i,j,k}}$ , is defined as the element-wise variance, reflecting the variability in  $X_{i,j,k}$  across different instances. Similarly, the element-wise mean,  $\mu_{X_{i,j,k}}$ , represents the average value of  $X_{i,j,k}$ . Formally, the element-wise mean and variance are defined as follows:

$$\mu_{X_{i,j,k}} = \mathbb{E}[X_{i,j,k}]$$
  
$$\sigma_{X_{i,i,k}}^2 = \mathbb{E}[(X_{i,j,k} - \mu_{X_{i,j,k}})^2].$$

**Element-wise covariance and the correlation coefficient**: To assess the relationship between corresponding elements in tensors *X* and *Y*, the element-wise covariance between corresponding elements in *X* and *Y* is calculated as follows:

$$Cov(X_{i,j,k}, Y_{i,j,k}) = \mathbb{E}[(X_{i,j,k} - \mu_{X_{i,j,k}})(Y_{i,j,k} - \mu_{Y_{i,j,k}})].$$

In turn, the (Pearson's) correlation coefficient between the corresponding elements in *X* and *Y* is given by

$$\operatorname{Corr}(X_{i,j,k}, Y_{i,j,k}) = \frac{\operatorname{Cov}(X_{i,j,k}, Y_{i,j,k})}{\sigma_{X_{i,i,k}}\sigma_{Y_{i,i,k}}},$$

where  $\sigma_{X_{i,j,k}}$  and  $\sigma_{Y_{i,j,k}}$  are the standard deviations of  $X_{i,j,k}$  and  $Y_{i,j,k}$ , respectively. This correlation is a standardized measure of a linear relationship that ranges between -1 and 1.

Given that each  $Z_{i,j,k}$  is the sum of  $X_{i,j,k}$  and  $Y_{i,j,k}$ , the element-wise variance in  $Z_{i,j,l}$  can be expressed as follows:

$$\sigma_{Z_{i,j,k}}^2 = \sigma_{X_{i,j,k}}^2 + \sigma_{Y_{i,j,k}}^2 + 2\text{Cov}(X_{i,j,k}, Y_{i,j,k}).$$

We can also synthesize these element-wise variances in the tensors into overall measures to capture general trends in these statistics across a representation tensor. This can be achieved by computing the averaged element-wise variance and covariance across all elements in tensor Z as follows:

$$\overline{\sigma_{Z}^{2}} = \frac{1}{h \cdot w \cdot c} \sum_{i=1,j=1,k=1}^{h,w,c} \left( \sigma_{X_{i,j,k}}^{2} + \sigma_{Y_{i,j,k}}^{2} + 2 \cdot \operatorname{Cov}(X_{i,j,k}, Y_{i,j,k}) \right)$$

$$= \frac{1}{h \cdot w \cdot c} \left( \sum_{i=1,j=1,k=1}^{h,w,c} \sigma_{X_{i,j,k}}^{2} + \sum_{i=1,j=1,k=1}^{h,w,c} \sigma_{Y_{i,j,k}}^{2} + 2 \sum_{i=1,j=1,k=1}^{h,w,c} \operatorname{Cov}(X_{i,j,k}, Y_{i,j,k}) \right)$$

$$= \overline{\sigma_{X}^{2}} + \overline{\sigma_{Y}^{2}} + 2\overline{\operatorname{Cov}(X,Y)},$$
(8)

where  $\sigma_X^2$  and  $\sigma_Y^2$  are the averaged element-wise variances in all individual elements in tensors *X* and *Y*, respectively, and  $\overline{\text{Cov}(X, Y)}$  is the averaged element-wise covariance of the corresponding elements across tensor *X* and *Y*.

Likewise, the averaged element-wise correlation coefficient is defined as follows:

$$\overline{\operatorname{Corr}(X,Y)} = \frac{1}{h \cdot w \cdot c} \sum_{i=1,j=1,k=1}^{h,w,c} \overline{\operatorname{Corr}(X_{i,j,k},Y_{i,j,k})}$$

Substituting *X* with  $Z_{l,F}$ , the functional component, and *Y* with  $Z_I$ , the informative component, gives us  $Z = Z_l$ , the representation of the sub-network  $h_{\theta_{1,l}}$  with identity shortcuts.

**Dynamics of the element-wise statistics**: Given the same dataset, the representations generated by a neural network may change as the parameters are updated through the learning algorithm. We aim to study how the statistics—the averaged element-wise variance and the averaged element-wise correlation coefficient—evolve as the network undergoes training. The dynamics of the element-wise statistics will be analyzed as a function of the training checkpoints.

More specifically, having decomposed the output of the sub-network with identity shortcuts  $Z_l$  into a functional component  $Z_{l;F}$  and an informative component  $Z_I$ , we introduce the notations  $Z_l^{(t)}$ ,  $Z_{l;F}^{(t)}$ , and  $Z_I^{(t)}$  to represent the values obtained from the neural network at checkpoint *t*. The subsequent sections will analyze how the aforementioned element-wise statistics evolve as a function of *t*.

## 5.2.2. Results and Analysis

**The iResCNN**: First, we analyze the dynamics of the element-wise correlation coefficient and the element-wise variance in the iResCNN. The results are displayed in Figure 9.



**Figure 9.** Dynamics of averaged element-wise correlation coefficients and averaged element-wise variance in the iResCNN: In the upper row, the curves in a darker color and the left axis show the dynamics of  $\overline{\text{Corr}(Z_{l;F}, Z_I)}$ , while the lighter curves and the right axis show the Z-X measure  $(m_{Z_l;X})$  obtained from Section 4.2. In the lower row, the dynamics of  $\overline{\sigma_{Z_l}^2}$ ,  $\overline{\sigma_{Z_{l;F}}^2}$ , and  $\overline{\sigma_{Z_l}^2}$  are visualized. The left, middle, and right panels show the representations of different modules. *l* is the index for the modules in the iResCNN shown in the right panel of Figure 3. Panels in the same row or column share the same axes.

The upper row in Figure 9 shows that  $\overline{\operatorname{Corr}(Z_{l;F}, Z_I)}$  is closely correlated with the Z-X dynamics (plotted with  $m_{Z_l;X}$ ), which reflect the magnitude of compression. More specifically, from epoch 0 to epoch 15,  $\overline{\operatorname{Corr}(Z_{l;F}, Z_l)}$  decreases as  $m_{Z_l;X}$  increases. However, from the 15th epoch onwards, the compression tends to be less pronounced as  $m_{Z_l;X}$  decreases mildly, while  $\overline{\operatorname{Corr}(Z_{l;F}, Z_I)}$  also increases after the 15th epoch. Additionally, the deeper module (e.g., l = 3) exhibits lower  $\overline{\operatorname{Corr}(Z_{l;F}, Z_I)}$ , corresponding to higher  $m_{Z_l;X}$ .

From the lower row in Figure 9, we observe that the averaged element-wise variance in the functional component  $\overline{\sigma_{Z_{l;F}}^2}$  is higher in deeper layers and tends to increase as the network exhibits compression (referencing the upper row in Figure 9). Notably, in the last module (l = 3),  $\overline{\sigma_{Z_{l;F}}^2}$  exceeded  $\overline{\sigma_{Z_l}^2}$ . This suggests that the functional components  $Z_{l;F}$  play an increasingly important role in the overall representation  $Z_l$  as the network deepens and exhibits compression.

Additionally, we find that the trends in  $\overline{\sigma_{Z_{l;F}}^2}$  seem to be negatively correlated with  $\overline{\sigma_{Z_{l}}^2}$ , consistent with the relationship shown in Equation (8) when substituting *X*, *Y*, and *Z* with *Z*<sub>1</sub>, *Z*<sub>1</sub>, *E*, and *Z*<sub>1</sub>, respectively. More specifically, Equation (8) states

$$\overline{\sigma_{Z_l}^2} = \overline{\sigma_{Z_I}^2} + \overline{\sigma_{Z_{l;F}}^2} + 2\overline{\operatorname{Cov}(Z_I, Z_{l;F})}.$$

Here,  $\overline{\sigma_{Z_l}^2}$  is constant. If  $\overline{\sigma_{Z_l}^2}$  decreases while  $\overline{\sigma_{Z_{l;F}}^2}$  increases, this implies a decrease in  $\overline{\text{Cov}(Z_I, Z_{l;F})}$ . In fact,  $\overline{\text{Cov}(Z_I, Z_{l;F})}$  has become negative, as indicated by the negative  $\text{Corr}(Z_l; Z_{l;F})$ , which essentially is a normalized measure of the covariance.

Finally,  $\overline{\sigma_{Z_{l,F}}^2}$  and  $\overline{\sigma_{Z_l}^2}$  are roughly on the same scale as  $\overline{\sigma_{Z_1}^2}$ , indicating that the representation is not dominated by either the functional or informative components. This contrasts with the trends observed in the ViT and MLP-Mixer, which will be discussed later.

These patterns suggest that the pronounced compression in the iResCNN may be due to the network learning to make the functional components more negatively correlated with the informative components, implying a potential canceling effect.

To further validate this conjecture, we examined histograms of the element-wise correlation coefficient between the functional and informative components across different modules and network epochs, as shown in Figure 10.



**Figure 10.** Histograms of element-wise correlation coefficients in the iResCNN: These histograms summarize the element-wise coefficients  $Corr(Z_{l;F;i}, Z_{I;i})$ , where *i* indexes the entries of the representation components.

The histograms corroborate our conjectures. At initialization (epoch 0), the elements in the functional and informative components can be either positively or negatively correlated, with the element-wise correlation coefficients spread from -1 to 1. As training progresses (from epoch 1 to epoch 10), an increasing number of elements show negative correlations. By epoch 10, when the network exhibits considerable compression (refer to the upper row in Figure 9, to the curves in a lighter color with the right axis), a significant portion of the elements displays a correlation close to -1. At the end of training, the level of compression is less pronounced compared to that at the 10th epoch, with the correlation coefficients for most elements showing a slight deviation away from -1.

Comparing different layers, the elements in the representations of the deeper layers show stronger negative correlations, indicating a more significant canceling effect.

Our observations support the hypothesis that canceling effects in the iResCNN, which we identified as a potential source of compression in the representations generated by subnetworks with identity shortcuts, play a crucial role. Conventionally, identity shortcuts are used to facilitate the training of very deep neural networks. However, our results suggest that these networks may inherently discover a mechanism for compression through cancelneural networks.

lation. This finding is significant, as compression is crucial for enhancing generalization in

**The MLP-Mixer**: We conducted the same experiments on the MLP-Mixer. In particular, we chose to present the last three residual modules, as they exhibit the most pronounced compression.

In the fourth module of the MLP-Mixer (l = 4), the elements in the functional components tend to be negatively correlated with the informative component, and as the representation exhibits more pronounced compression, the correlation decreases. This pattern is consistent with what was observed in the iResCNN, implying a canceling effect.

However, in the fifth (l = 5) and sixth (l = 6) modules, only in the first five epochs do the averaged correlation coefficients decrease as compression increases. In later epochs, while the Z-X measure continues to increase, the correlation coefficient stops decreasing and stabilizes at a certain negative value.

Referring to the lower row in Figure 11, it is evident that the averaged element-wise variance in the summed output  $Z_l$  is dominated by the functional component  $Z_{l;F}$ , as  $\overline{\sigma_{Z_l}^2}$  closely aligns with  $\overline{\sigma_{Z_{l;F}}^2}$ , with both being significantly higher in scale than  $\overline{\sigma_{Z_l}^2}$ . This suggests that the informative components may be overwhelmed by the functional components, potentially complicating the reconstruction of the input data from the summed output. Using a neural-network-based estimator to estimate the MMSE, the dominance of the functional components may bias the estimation of the true MMSE, resulting in a positive Z-X measure. Nevertheless, this observation provides insight into why reconstructing the input data from the representation becomes increasingly difficult: the variance in the representation is dominated by the functional components, which are processed by neural network layers.



**Figure 11.** Dynamics of the averaged element-wise correlation coefficients and averaged element-wise variance in the MLP-Mixer: In the upper row, the curves in a darker color and the left axis show the dynamics of  $\overline{\text{Corr}(Z_{l;F}, Z_I)}$ , while the lighter curves and the right axis show the Z-X measure  $(m_{Z_l;X})$  obtained from Section 4.4. In the lower row, the dynamics of  $\overline{\sigma_{Z_l}^2}$ ,  $\overline{\sigma_{Z_{l;F}}^2}$ , and  $\overline{\sigma_{Z_l}^2}$  are visualized. The left, middle, and right panels show the representations of different modules. *l* is the index for the modules in the MLP-Mixer shown in the right panel in Figure 6. Panels in the same row or column share the same axes.

In contrast, this dominance of  $Z_{l;F}$  in  $\sigma_{Z_l}^2$  was not observed in the iResCNN. More specifically, the lower row in Figure 9 shows that  $\sigma_{Z_{l;F}}^2$  and  $\sigma_{Z_l}^2$  are on the same scale as  $\sigma_{Z_l}^2$ , and the variance in  $\sigma_{Z_{l;F}}^2$  decreases compared to that of  $\sigma_{Z_l}^2$ , rather than them evolving in synchronization.

The ViT: Finally, we examine the ViT model, focusing on the dynamics of averaged the element-wise covariance, Z-X measure, and averaged element-wise variance, as illustrated in Figure 12. Particular attention is given to the last three MHSA modules in the ViT, which exhibit the most pronounced compression phases.



**Figure 12.** Dynamics of averaged statistics for the ViT: In the upper row, the curves in a darker color and the left axis show the dynamics of  $\overline{\text{Corr}(Z_{l;F}, Z_I)}$ , while the lighter curves and the right axis show the Z-X measure  $(m_{Z_l;X})$  obtained from Section 4.4. In the lower row, the dynamics of  $\overline{\sigma_{Z_l}^2}$ ,  $\overline{\sigma_{Z_{l;F}}^2}$ , and  $\overline{\sigma_{Z_l}^2}$  are visualized. The left, middle, and right panels show the representations of different modules. *l* is the index for the modules in the ViT shown in the left panel in Figure 6. Panels in the same row or column share the same axes.

We can observe from Figure 12 that the fourth, fifth, and sixth MHSA modules display trends similar to those in the sixth module of the token mixer in the MLP-Mixer. Specifically,  $\overline{\text{Corr}(Z_{l;F}, Z_I)}$  decreases during the initial five epochs, implying a canceling effect, and then increases rapidly and eventually becomes positive during subsequent epochs.

Meanwhile,  $\overline{\sigma_{Z_l}^2}$  and  $\overline{\sigma_{Z_{l;F}}^2}$  increase significantly from the fifth epoch onwards, surpassing  $\overline{\sigma_{Z_l}^2}$  in magnitude. The trends in  $\overline{\sigma_{Z_l}^2}$  and  $\overline{\sigma_{Z_{l;F}}^2}$  are also well synchronized.

These observations suggest that the initial compression phase is driven by a canceling effect. However, as the network is trained over more epochs, the element-wise variances in the functional components overwhelm those of the informative components, indicating a shift in the network's internal dynamics.

Both the MLP-Mixer and the ViT make a point similar to that for the iResCNN and conventional feed-forward neural networks investigated in [17], where the networks discover a compression mechanism that may be useful for generalization (despite the fact that they use shortcuts to improve training). This, in effect, means that shortcuts do not impair generalization.

#### 6. Conclusions

This paper focuses on neural networks with additive shortcuts, examining their fitting and compression phenomena in learning dynamics. Empirical experiments show that models with non-identity shortcuts (NISs), such as the ResCNN, exhibit an initial fitting phase followed by a subsequent compression phase, similar to neural networks without additive shortcuts examined in the prior literature. However, in neural networks with identity shortcuts (ISs), such as the iResCNN, ViT, and MLP-Mixer, the initial fitting phase is notably absent, and these models instead display pronounced compression phases. Further analysis in this study investigates the mechanisms underlying the observed behaviors in models with ISs. We conjecture and empirically validate that the absence of the initial fitting phase may be due to the ability of ISs to forward sufficient information about the ground-truth labels even at random initialization of the subject network.

Additionally, the pronounced compression phase in models with ISs appears to arise from different underlying mechanisms. For the iResCNN, the compression throughout training primarily results from a cancellation effect within its functional components. In ViTs and MLP-Mixers, the situation is more complex. Initially, the functional components also exhibit cancellation effects on the informative components; however, as the training progresses, the functional components increasingly dominate, overwhelming the informative components and resulting in significant data compression. These insights are crucial, as they deepen our understanding of the compression mechanisms in networks with identity shortcuts and suggest potential strategies, such as compression-oriented regularization, to enhance the network efficiency and performance.

## 6.1. Limitations and Future Work

While this study provides valuable insights into information bottleneck dynamics in networks with additive shortcuts, it has certain limitations that should be addressed in future work:

- The dataset and task scope: This study uses a single dataset (CIFAR-10) and does not examine other data types or tasks, such as language or video processing, which may exhibit different IB dynamics. Testing these findings across larger and more diverse datasets, especially those with varied data types, is essential to determine the generality of these dynamics.
- Inductive bias in the ViT and MLP-Mixer: As noted in the comments, the ViT and MLP-Mixer models are known to have weaker inductive biases than convolutional models, making them less optimal for small datasets like CIFAR-10 when they are trained from scratch. While we trained the ViT and MLP-Mixer to competitive levels of 86.14% and 83.76% accuracy, respectively, these results are still lower than the benchmarks achieved by convolutional models like ResNet. This limitation highlights the need for testing on larger datasets or using pre-trained versions of these models to capture the shortcut path dynamics better.
- Shortcut configurations: This study focuses on additive shortcuts and does not explore alternative configurations, such as the concatenation shortcuts used in DenseNet or combinations of identity and non-identity shortcuts. Investigating these different shortcut designs may reveal additional dynamics that influence IB behavior in novel ways.
- Layer-specific analysis: The compression behavior in IS-based models appears to vary across layers, suggesting that layer-wise analysis could yield more precise insights into IB dynamics. Future studies could benefit from examining shortcut path dynamics at the layer-specific level to identify potential opportunities to fine-tune the shortcut configurations and improve performance.
- Memory constraints and computational intensity: Due to the memory-intensive nature of the ViT and MLP-Mixer, our study required substantial memory resources to implement multiple estimations across layers, checkpoints, and measures (Z-X and Z-Y). Convolutional networks are comparatively more memory-efficient due to shared kernel use and pooling operations, which reduce the memory intensity. In contrast, ViTs' tokenized representation and dense matrix multiplications increase the computational demands, making these models more challenging to study. Future work could explore memory-optimized methods for self-attention or alternative ways to handle large-scale estimations.

#### 6.2. Summary

In summary, this paper extends the landscape of information bottleneck dynamics by exploring models with additive shortcuts and their unique learning phenomena. Despite using a single dataset, the patterns observed align with IB theory and the related literature, suggesting that they may generalize to broader contexts. Future work could build on this foundation by addressing the limitations noted here, testing these dynamics across more datasets, configurations, and applications to deepen our understanding of neural networks with additive shortcuts.

Author Contributions: Conceptualization, Z.L. and M.R.D.R.; methodology, Z.L.; software, Z.L.; validation, Z.L.; formal analysis, Z.L.; writing—original draft preparation, Z.L.; writing—review and editing, M.R.D.R.; supervision, M.R.D.R. All authors have read and agreed to the published version of the manuscript.

Funding: The first author is funded by China Scholarship Council and UCL Dean's Prize.

Institutional Review Board Statement: Not applicable.

**Data Availability Statement:** The data presented in this study are available on request from the corresponding author.

Conflicts of Interest: The authors declare no conflicts of interest.

#### References

- 1. LeCun, Y.; Bengio, Y.; Hinton, G. Deep learning. Nature 2015, 521, 436–444. [CrossRef] [PubMed]
- 2. LeCun, Y.; Bengio, Y. Convolutional networks for images, speech, and time series. *Handb. Brain Theory Neural Netw.* **1995**, 3361, 1995.
- 3. Krizhevsky, A.; Sutskever, I.; Hinton, G.E. Imagenet classification with deep convolutional neural networks. *Adv. Neural Inf. Process. Syst.* **2012**, *25*, 84–90. [CrossRef]
- He, K.; Zhang, X.; Ren, S.; Sun, J. Deep residual learning for image recognition. In Proceedings of the IEEE Conference on Computer Vision and Pattern Recognition, Las Vegas, NV, USA, 27–30 June 2016; pp. 770–778.
- 5. Vaswani, A. Attention is all you need. Adv. Neural Inf. Process. Syst. 2017, 30, 6000–6010.
- Grigorescu, S.; Trasnea, B.; Cocias, T.; Macesanu, G. A survey of deep learning techniques for autonomous driving. *J. Field Robot.* 2020, 37, 362–386. [CrossRef]
- 7. Elsken, T.; Metzen, J.H.; Hutter, F. Neural architecture search: A survey. J. Mach. Learn. Res. 2019, 20, 1–21.
- 8. Tan, M.; Le, Q.V. EfficientNet: Rethinking Model Scaling for Convolutional Neural Networks. arXiv 2019, arXiv:1905.11946.
- 9. LeCun, Y.; Bottou, L.; Bengio, Y.; Haffner, P. Gradient-based learning applied to document recognition. *Proc. IEEE* **1998**, *86*, 2278–2324. [CrossRef]
- 10. Hochreiter, S. Long Short-term Memory. In Neural Computation; MIT-Press: Cambridge, MA, USA, 1997.
- 11. Dosovitskiy, A. An image is worth  $16 \times 16$  words: Transformers for image recognition at scale. *arXiv* **2020**, arXiv:2010.11929.
- 12. Tolstikhin, I.O.; Houlsby, N.; Kolesnikov, A.; Beyer, L.; Zhai, X.; Unterthiner, T.; Yung, J.; Steiner, A.; Keysers, D.; Uszkoreit, J.; et al. Mlp-mixer: An all-mlp architecture for vision. *Adv. Neural Inf. Process. Syst.* **2021**, *34*, 24261–24272.
- Saxe, A.M.; Bansal, Y.; Dapello, J.; Advani, M.; Kolchinsky, A.; Tracey, B.D.; Cox, D.D. On the information bottleneck theory of deep learning. *J. Stat. Mech. Theory Exp.* 2019, 2019, 124020. [CrossRef]
- 14. Tishby, N.; Pereira, F.C.; Bialek, W. The information bottleneck method. arXiv 2000, arXiv:physics/0004057.
- Tishby, N.; Zaslavsky, N. Deep learning and the information bottleneck principle. In Proceedings of the 2015 IEEE Information Theory Workshop (ITW), Jeju Island, Republic of Korea, 11–15 October 2015; pp. 1–5.
- 16. Shwartz-Ziv, R.; Tishby, N. Opening the black box of deep neural networks via information. arXiv 2017, arXiv:1703.00810.
- 17. Lyu, Z.; Aminian, G.; Rodrigues, M.R. On Neural Networks Fitting, Compression, and Generalization Behavior via Information-Bottleneck-like Approaches. *Entropy* **2023**, *25*, 1063. [CrossRef]
- 18. Beyer, L.; Zhai, X.; Kolesnikov, A. Better plain ViT baselines for ImageNet-1k. arXiv 2022, arXiv:2205.01580.
- Li, Z.; Wallace, E.; Shen, S.; Lin, K.; Keutzer, K.; Klein, D.; Gonzalez, J. Train big, then compress: Rethinking model size for efficient training and inference of transformers. In Proceedings of the International Conference on Machine Learning, Virtual Event, 13–18 July 2020; pp. 5958–5968.
- 20. Bai, Y.; Mei, J.; Yuille, A.L.; Xie, C. Are transformers more robust than cnns? Adv. Neural Inf. Process. Syst. 2021, 34, 26831–26843.
- 21. Alammar, J. The Illustrated Transformer. 2024. Available online: https://jalammar.github.io/illustrated-transformer/ (accessed on 9 October 2024).
- Liu, Z.; Lin, Y.; Cao, Y.; Hu, H.; Wei, Y.; Zhang, Z.; Lin, S.; Guo, B. Swin transformer: Hierarchical vision transformer using shifted windows. In Proceedings of the IEEE/CVF International Conference on Computer Vision, Montreal, BC, Canada, 10–17 October 2021; pp. 10012–10022.

- Wu, K.; Peng, H.; Chen, M.; Fu, J.; Chao, H. Rethinking and improving relative position encoding for vision transformer. In Proceedings of the IEEE/CVF International Conference on Computer Vision, Montreal, BC, Canada, 10–17 October 2021; pp. 10033–10041.
- 24. Kazemnejad, A.; Padhi, I.; Natesan Ramamurthy, K.; Das, P.; Reddy, S. The impact of positional encoding on length generalization in transformers. *Adv. Neural Inf. Process. Syst.* 2024, *36*, 24892–24928.
- 25. Ke, G.; He, D.; Liu, T.Y. Rethinking positional encoding in language pre-training. arXiv 2020, arXiv:2006.15595.
- Chen, P.C.; Tsai, H.; Bhojanapalli, S.; Chung, H.W.; Chang, Y.W.; Ferng, C.S. A simple and effective positional encoding for transformers. *arXiv* 2021, arXiv:2104.08698.
- Leem, S.; Seo, H. Attention Guided CAM: Visual Explanations of Vision Transformer Guided by Self-Attention. In Proceedings of the AAAI Conference on Artificial Intelligence, Vancouver, BC, Canada, 20–27 February 2024; Volume 38, pp. 2956–2964.
- 28. Ma, J.; Bai, Y.; Zhong, B.; Zhang, W.; Yao, T.; Mei, T. Visualizing and understanding patch interactions in vision transformer. *IEEE Trans. Neural Netw. Learn. Syst.* 2023, *35*, 13671–13680. [CrossRef]
- Huang, G.; Liu, Z.; Van Der Maaten, L.; Weinberger, K.Q. Densely connected convolutional networks. In Proceedings of the IEEE Conference on Computer Vision and Pattern Recognition, Honolulu, HI, USA, 21–26 July 2017; pp. 4700–4708.
- Wang, P. Lucidrains/MLP-Mixer-Pytorch: An All-MLP Solution for Vision, from Google Ai. 2022. Available online: https://github. com/lucidrains/mlp-mixer-pytorch (accessed on 16 October 2023).
- Zeiler, M.D.; Fergus, R. Visualizing and understanding convolutional networks. In Proceedings of the Computer Vision–ECCV 2014: 13th European Conference, Zurich, Switzerland, 6–12 September 2014; Proceedings, Part I 13; Springer: Berlin/Heidelberg, Germany, 2014; pp. 818–833.
- Chattopadhyay, A.; Sarkar, A.; Howlader, P.; Balasubramanian, V.N. Grad-CAM++: Generalized Gradient-Based Visual Explanations for Deep Convolutional Networks. In Proceedings of the 2018 IEEE Winter Conference on Applications of Computer Vision (WACV), Lake Tahoe, NV, USA, 12–15 March 2017; pp. 839–847.
- 33. Selvaraju, R.R.; Das, A.; Vedantam, R.; Cogswell, M.; Parikh, D.; Batra, D. Grad-CAM: Visual Explanations from Deep Networks via Gradient-Based Localization. *Int. J. Comput. Vis.* **2016**, *128*, 336–359. [CrossRef]
- Xu, L.; Yan, X.; Ding, W.; Liu, Z. Attribution rollout: A new way to interpret visual transformer. J. Ambient. Intell. Humaniz. Comput. 2023, 14, 163–173. [CrossRef]
- 35. Abnar, S.; Zuidema, W. Quantifying attention flow in transformers. *arXiv* **2020**, arXiv:2005.00928.
- Vapnik, V.N.; Chervonenkis, A.Y. On the uniform convergence of relative frequencies of events to their probabilities. In *Measures of Complexity: Festschrift for Alexey Chervonenkis*; Springer: Berlin/Heidelberg, Germany, 2015; pp. 11–30.
- 37. Waltz, D. A Theory of the Learnable. Commun. ACM 1984, 27, 1134–1142.
- 38. Shalev-Shwartz, S.; Ben-David, S. *Understanding Machine Learning: From Theory to Algorithms*; Cambridge University Press: Cambridge, UK, 2014.
- Neyshabur, B.; Bhojanapalli, S.; McAllester, D.A.; Srebro, N. A PAC-Bayesian Approach to Spectrally-Normalized Margin Bounds for Neural Networks. arXiv 2017, arXiv:1707.09564.
- Li, H.; Xu, Z.; Taylor, G.; Studer, C.; Goldstein, T. Visualizing the loss landscape of neural nets. Adv. Neural Inf. Process. Syst. 2018, 31, 6391–6401.
- 41. Alemi, A.A.; Fischer, I.; Dillon, J.V.; Murphy, K. Deep variational information bottleneck. arXiv 2016, arXiv:1612.00410.
- 42. Polyanskiy, Y.; Wu, Y. Lecture notes on information theory. Lect. Notes ECE563 (UIUC) 2014, 6, 7.
- Goldfeld, Z.; van den Berg, E.; Greenewald, K.; Melnyk, I.; Nguyen, N.; Kingsbury, B.; Polyanskiy, Y. Estimating information flow in deep neural networks. arXiv 2018, arXiv:1810.05728.
- 44. Darlow, L.N.; Storkey, A. What information does a ResNet compress? arXiv 2020, arXiv:2003.06254.
- Noshad, M.; Zeng, Y.; Hero, A.O. Scalable mutual information estimation using dependence graphs. In Proceedings of the ICASSP 2019–2019 IEEE International Conference on Acoustics, Speech and Signal Processing (ICASSP), Brighton, UK, 12–17 May 2019; pp. 2962–2966.
- 46. Jónsson, H.; Cherubini, G.; Eleftheriou, E. Convergence behavior of DNNs with mutual-information-based regularization. *Entropy* **2020**, *22*, 727. [CrossRef] [PubMed]
- Kirsch, A.; Lyle, C.; Gal, Y. Scalable training with information bottleneck objectives. In Proceedings of the International Conference on Machine Learning (ICML): Workshop on Uncertainty and Robustness in Deep Learning, Virtual, 17–18 July 2020; pp. 17–18.
- Krizhevsky, A.; Hinton, G. Learning Multiple Layers of Features from Tiny Images. 2009. Available online: https://xueshu.baidu. com/usercenter/paper/show?paperid=c55665fb879e98e130fce77052d4c8e8 (accessed on 29 October 2024).
- LeCun, Y.; Cortes, C.; Burges, C. MNIST handwritten digit database. ATT Labs 2010, 2. Available online: http://yann.lecun.com/ exdb/mnist (accessed on 29 October 2024).
- 50. Loshchilov, I.; Hutter, F. Sgdr: Stochastic gradient descent with warm restarts. arXiv 2016, arXiv:1608.03983.
- 51. Carratino, L.; Ciss'e, M.; Jenatton, R.; Vert, J.P. On Mixup Regularization. J. Mach. Learn. Res. 2020, 23, 325:1–325:31.
- Yun, S.; Han, D.; Oh, S.J.; Chun, S.; Choe, J.; Yoo, Y.J. CutMix: Regularization Strategy to Train Strong Classifiers with Localizable Features. In Proceedings of the 2019 IEEE/CVF International Conference on Computer Vision (ICCV), Seoul, Republic of Korea, 27 October–2 November 2019; pp. 6022–6031.
- 53. Szegedy, C.; Vanhoucke, V.; Ioffe, S.; Shlens, J.; Wojna, Z. Rethinking the Inception Architecture for Computer Vision. In Proceedings of the 2016 IEEE Conference on Computer Vision and Pattern Recognition (CVPR), Las Vegas, NV, USA, 27–30 June 2016; pp. 2818–2826.

- 54. Kentaro, Y. Kentaroy47/Vision-Transformers-CIFAR10: Let's Train Vision Transformers (VIT) for CIFAR 10! 2023. Available online: https://github.com/kentaroy47/vision-transformers-cifar10 (accessed on 16 October 2023).
- 55. Dumoulin, V.; Visin, F. A guide to convolution arithmetic for deep learning. *arXiv* 2016, arXiv:1603.07285.

**Disclaimer/Publisher's Note:** The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.